\documentstyle[12pt]{article}
\newtheorem{thm}{Theorem}
\newcommand{\inter}{{\rm int\,}}
\newcommand{\cl}{{\rm cl\,}}
\begin{document}
\begin{center}

{\large\bf C\&O 367 \\ 
Assignment 3 (Iterative Methods for Unconstrained Minimization)
}
\end{center}
\begin{flushleft}
{\large  Due on Thursday, Mar. 4, 1999.   Instructor H. Wolkowicz
}
\end{flushleft}
\bigskip

\begin{enumerate}
\item (10)
Page 129, problems \# 3 and \# 4.
\item (10)
Page 129, use MATLAB (e.g. fminu) to find the minimum of the 
function $f$ given in problem  \# 4.
(Please include some details of each iteration.)
\item (10)
Page 130, \# 8, \# 11, \# 12
\end{enumerate}
\end{document}
